Author's response to reviews

Title: The CANadian Pediatric Weight management Registry (CANPWR): Study Protocol

Authors:

Katherine M Morrison (kmorrison@mcmaster.ca)
Samah Damanhoury (Samah.damanhoury@hotmail.com)
Annick Buchholz (Buchholz@cheo.on.ca)
Jean Pierre Chanoine (jchanoine@cw.bc.ca)
Marie Lambert (mlambert@ssss.gouv.qc.ca)
Mark S Tremblay (mtremblay@cheo.on.ca)
Glenn Berall (Glenn.Berall@nygh.on.ca)
Jill Hamilton (jill.hamilton@sickkids.ca)
Anne Marie Laberge (anne-marie.laberge.hsj@ssss.gouv.qc.ca)
Laurent Legault (Laurent.legault@muhc.mcgill.ca)
Lehana Thabane (ThabanL@mcmaster.ca)
Monica Jakymyshyn (monica.jakymyshyn@phri.ca)
Kathryn A Ambler (kambler@med.ualberta.ca)
Geoff DC Ball (gdball@ualberta.ca)

Version: 2 Date: 14 May 2014

Author's response to reviews:

In addition to the attached cover letter outlining that the ethics approvals and funding documentation has been sent to the appropriate email address, we have added line numbers to the revised manuscript.